Author's response to reviews

Title: Fibroblast activation protein-alpha-expressing fibroblasts promote the progression of pancreatic ductal adenocarcinoma

Authors:

Tomoya Kawase (oneworld_traveller@yahoo.co.jp)
Yumiko Yasui (yumikoy@med.kawasaki-m.ac.jp)
Sohji Nishina (018ep@med.kawasaki-m.ac.jp)
Yuichi Hara (yuich_h777@med.kawasaki-m.ac.jp)
Izumi Yanatori (yanatori@med.kawasaki-m.ac.jp)
Yasuyuki Tomiyama (tomiyama@med.kawasaki-m.ac.jp)
Yoshihiro Nakashima (makazi777@med.kawasaki-m.ac.jp)
Koji Yoshida (kojiyos@med.kawasaki-m.ac.jp)
Fumio Kishi (kishi@med.kawasaki-m.ac.jp)
Masafumi Nakamura (mnakamura@med.kawasaki-m.ac.jp)
Keisuke Hino (khino@med.kawasaki-m.ac.jp)

Version: 4 Date: 31 July 2015

Author's response to reviews: see over
Dear Editors,

I sincerely thank you for the decision to accept our manuscript for publication in BMC Gastroenterology.

The requested changes by Editorial office for proceeding to the next process have been made as described below.

Editorial request:

1.) Please remove the short running title.
   I have removed the short running title.

2.) Please place the abstract on a separate page to the title page.
   I have placed the abstract on a separate page to the title page.

3.) Please provide the full name of the IRB when stating ethical approval.
   I have provided the full name of the IRB when stating ethical approval. (Line 12 in page 6)

4.) Please confirm whether informed consent was obtained from all patients. If the need for informed consent was waived by the IRB, please clearly state this.
   I have clarified that the need for informed consent was waived by the Research Ethics Committee. (Lines 12-14 in page 6)

5.) Please provide a Conclusions section at the end of the Discussion.
   I have provided a Conclusions section at the end of the Discussion. (Lines 1-6 in page 17)